Author’s response to reviews

Title: The TXA2R rs1131882, P2Y1 rs1371097 and GPⅢa rs2317676 three-loci interactions may increase the risk of carotid stenosis in patients with ischemic stroke

Authors:

Jing Lin (22350277@qq.com)
Qiang Zhou (zhouqiang5613@163.com)
Xingyang Yi (1842942576@qq.com)
Ruyue Huang (zbyhry@163.com)
Zhenxiao Chai (yixingyang64@163.com)

Version: 1 Date: 05 Aug 2018

Author’s response to reviews:

August 05, 2018

Editorial Office,

BMC Neurology

Dear Editor,

We would like to re-submit our manuscript, “The TXA2R rs1131882, P2Y1 rs1371097 and GPⅢa rs2317676 three-loci interactions may increase the risk of carotid stenosis in Chinese population”, by Xingyang Yi, and Jing Lin et al. for consideration for publication in BMC Neurology.

We very thank the editors for their comments and suggestions. Accordingly, we have revised the manuscript.
This paper has not been published elsewhere in whole or in part. All authors have read and approved the content, and agree to submit it for consideration for publication in your journal. There are no ethical/legal conflicts involved in the article.

Our manuscript should be of interest to BMC Neurology readers, and in particular, researchers who are interested in the underlying basic mechanisms of carotid stenosis. We also believe that more broad readers of public will appreciate our findings, and may conduct follow-up studies to replicate our findings and better clarify the identified relationship.

We hope that the revision is acceptable for the publication in your journal.

We look forward to hearing from you soon.

Yours sincerely,

Xingyang Yi and Jing Lin, MD
Professor of Neurology
Department of Neurology
The People's Hospital of Deyang City
No 173, North Taishan Road, Deyang 618000, Sichuan, China
E-mail: yixingyang64@126.com
Tel: 86-838-2204910; Fax: 86-838-2204910